Author's response to reviews

Title: Association of obesity with socioeconomic status among adults of ages 18 to 80 years in rural Northwest China

Authors:

Leilei Pei Dr (peileilei424@163.com)
Yue Cheng (chengy@mail.xjtu.edu.cn)
Yijun Kang (tikyi@mail.xjtu.edu.cn)
Shuyi Yuan (renlin88310@sina.com)
Hong Yan (yhpaper605@gmail.com)

Version: 7  Date: 3 February 2015

Author's response to reviews:

Dear Editor,

RE: MS: 4964834381303880-Association of obesity with socioeconomic status among adults of ages 18 to 80 years in rural Northwest China

We are really grateful to yours and other reviewers’ critical comments and thoughtful suggestions. We have considered these comments and suggestions carefully. All changes made to the text are marked in red in MS Word. We hope the new manuscript will meet your magazine’s standard. Below you will find our point-to-point responses to the reviewers’ comments/questions:

Reviewer: Lifoter Navti ’ comments

Minor Essential Revision

Abstract:

Line 45: Adjusted odds ratios were used........
Line 46: ..........after controlling for confounding factors using logistic regression

Reply: We are grateful for pointing out this errors. Accordingly, we have corrected the two sentences marked in red in the abstract.

Additional requirements:

The Competing Interests, Authors' Contributions and Acknowledgement sections are missing from the latest version of the manuscript. Please add these sections and please also ensure that your revised manuscript conforms to the journal style.

Reply: We are very sorry for this errors. In the end of the manuscript, the Competing Interests, Authors' Contributions and Acknowledgement sections were added. We hope the new manuscript will meet your magazine’s standard.
Yours sincerely,

Hong Yan
Institute: Department of Epidemiology and Health Statistics, School of Public Health, Xi’an Jiaotong University Health Science Center.
Address: No.76 West Yanta Road, Xi’an, Shaanxi 710061, P. R. China.
Tel: +86 29 8265 5001
Fax: +86 29 8265 5387
E-mail: yhpaper605@sina.com